Author's response to reviews

Title: Thyroid disease is a favorable prognostic factor in achieving sustained virologic response in chronic hepatitis C undergoing combination therapy: A nested case control study

Authors:  
Huy A Tran (huy.tran@hnehealth.nsw.gov.au)  
Robert Gibson (rgibson@hotmail.com)  
Glenn E M Reeves (greeves@yahoo.com)

Version: 2  Date: 5 October 2010

Author's response to reviews:

To the Editorial team:

This submission has been revised to include the Trial Registration number for the study. The number has been inserted into the last line of the abstract of the manuscript as requested.

With kind regards

HUY TRAN